Author's response to reviews

Title: Diagnostic validity of the MINI-KID disorder classifications in specialized child and adolescent psychiatric outpatient clinics in Sweden

Authors:
Camilla Högberg (camilla.hogberg@sll.se)
Eva Billstedt (eva.billstedt@gn.c.gu.se)
Caroline Björck (caroline.bjorck@sll.se)
Per-Olof Björck (pobjorck@hotmail.com)
Stephan Ehlers (stephan.ehlers@vgregion.se)
Lars-Henry Gustle (lars.henry.gustle@skane.se)
Clara Hellner (clara.hellner@ki.se)
Henrik Höök (henrik.hook@sll.se)
Eva Serlachius (eva.serlachius@ki.se)
Mats Svensson (mats.a.svensson@skane.se)
Jan-Olov Larsson (jan-olov.larsson@ki.se)

Version: 3 Date: 15 Apr 2019

Author’s response to reviews:

Your manuscript "Diagnostic validity of the MINI-KID disorder classifications in specialized child and adolescent psychiatric outpatient clinics in Sweden" (BPSY-D-18-00792R2) has been assessed by our reviewers. Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Psychiatry, once you have carried out some essential revisions.

1. Please rename the "Subjects and Methods" section so that it appears as 'Methods'.

The section is now renamed to Methods.

2. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in
the text, they should be defined in the text at first use and included in this list.

A list of abbreviations is now added after the Conclusions section. The abbreviations used in the text are now defined in the text at first use and included in the list.